
Monday 23 April 2012 
12th Scheduling Meeting

 
_______________

Minutes
_______________

Present : Mei Bai,  Bill  Christie,  Wolfram Fischer,  Xiaochun He,  Peter  Ingrassia,  Yun Luo, 
François Méot, Phil Pile.

 Brief overview of the coming week : see table next page

 Excerpts from the discussion :
- STAR and PHENIX do not need maintenance on next Wednesday, they are ready to keep on 

running instead. Question pending, though : 
we are still waiting for informations from LIPA – i.e., Paul - regarding whether LIPA 

would need to pull the plug on Wednesday. 
- U stores setting up requires further work at injection (re-bucketting), and on transition gamma. 

Once this is completed, then can go to Physics stores – could be tomorrow.  Next main development 
will be commissioning of stochastic cooling. 

- Question of low energy (5 GeV) runs addressed. 
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- UxU :  stores being delivered, beam set up on-going
- NSRL running
- BLIP running 

By the way : BLIP will operate one 
week longer - until July 31 - to meet 

isotope requirements.

Mon. 23

EBIS :  cryo-pump installation, started 0700am.
Maintenance behind this repair.

NSRL scheduled to run ~0900-1700 with p from LINAC.

EBIS : should end ~1000pm.
Includes RHIC access : RF and other.

RHIC could be restored ~0800pm

Tue. 24 UxU setup goes on.

Wed. 25
Maintenance could be cancelled - approved on PHENIX and 

STAR side.  Decision depends on projects at LIPA. 

If no maintenance, plan booster RF 
access behind store. If maintained, plan 

completing booster RF work before 
1100am so not to affect NSRL.

Thu. 26

Fri. 27

Sat. 28

Sun. 29

Mon. 30

NEXT SCHEDULING MEETING
MONDAY 30 APRIL


